Author’s response to reviews

Title: Depressive and Anxiety Symptoms and Associated factors among Postnatal Women in Enugu-North Senatorial District, South-East Nigeria: A Cross-sectional Study

Authors:

Olaoluwa Samson Agbaje (agbajesam@yahoo.com)
Joy Anyanwu (joy.ananywu@unn.edu.ng)
Prince Iheanachor Umoke (prince.umoke@unn.edu.ng)
Tochi Iwuagwu (tochi.iwuagu@unn.edu.ng)
Cylia Iweama (cylia.iweama@unn.edu.ng)
Eyuche Ozoemena (eyuche.ozoemena@unn.edu.ng)
Ijeoma Nnaji (cdccpubhealth2010@gmail.com)

Version: 4 Date: 03 Dec 2018

Author’s response to reviews:

1. We have explored associations in patients with both depression and anxiety. This is presented in Table 5.

2. We have interaction terms in the multinomial logistic model. This is presented in Table 6.

3. We have edited the cross-reference in the appendix using supplemental Tables.